Models of care for the delivery of secondary fracture prevention after hip fracture: a health service cost, clinical outcomes and cost-effectiveness study within a region of England

Andrew Judge, 1,2* M Kassim Javaid, 1,2 José Leal, 3 Samuel Hawley, 1 Sarah Drew, 1 Sally Sheard, 1 Daniel Prieto-Alhambra, 1,2,4 Rachael Gooberman-Hill, 5 Janet Lippett, 6 Andrew Farmer, 7 Nigel Arden, 1,2 Alastair Gray, 3 Michael Goldacre, 8 Antonella Delmestri 1 and Cyrus Cooper 1,2

¹Oxford National Institute for Health Research (NIHR) Musculoskeletal Biomedical Research Unit, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

²Medical Research Council (MRC) Lifecourse Epidemiology Unit, University of Southampton, Southampton General Hospital, Southampton, UK

³Health Economics Research Centre, Nuffield Department of Population Health, University of Oxford, Oxford, UK

⁴GREMPAL Research Group (IDIAP Jordi Gol) and Musculoskeletal Research Unit (Fundació IMIM-Parc Salut Mar), Universitat Autònoma de Barcelona, Barcelona, Spain

⁵School of Clinical Sciences, University of Bristol, Southmead Hospital, Bristol, UK ⁶Elderly Care Unit, Royal Berkshire Hospital, Reading, UK

⁷Nuffield Department of Primary Care Health Sciences, Oxford, UK

⁸Unit of Health Care Epidemiology, Nuffield Department of Population Health, University of Oxford, Oxford, UK

^{*}Corresponding author

Declared competing interests of authors: Daniel Prieto-Alhambra has received unrestricted research and educational grants from Amgen and Bioibérica S.A. Nigel Arden reports personal fees from Merck Sharp & Dohme (MSD), grants and personal fees from Roche, personal fees from Smith and Nephew, personal fees from Q-Med, personal fees from Nicox, personal fees from Flexion, personal fees from Bioibérica and personal fees from Servier. Cyrus Cooper has received consultancy fees, lecture fees and honoraria from Amgen, GlaxoSmithKline, Alliance for Better Bone Health, Eli Lilly, Pfizer, Novartis, MSD, Servier, Medtronic and Roche. M Kassim Javaid has in the last 5 years received honoraria for travel expenses, speaker fees and/or advisory committees from Lilly UK, Amgen, Servier, MSD, Medtronic, Internis, Consilient Health and Jarrow Formulas. He also serves on the Scientific Committee of the National Osteoporosis Society and International Osteoporosis Foundation. Andrew Judge has received consultancy fees, lecture fees and honoraria from Servier, UK Renal Registry, Oxford Craniofacial Unit, IDIAP Jordi Gol and Freshfields Bruckhaus Deringer, is a member of the Data Safety and Monitoring Board (which involved receipt of fees) from Anthera Pharmaceuticals, Inc., and received consortium research grants from Roche.

Published September 2016

DOI: 10.3310/hsdr04280

Plain English summary

Models of care for delivery of secondary fracture prevention

Health Services and Delivery Research 2016; Vol. 4: No. 28

DOI: 10.3310/hsdr04280

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

A total of 87,000 people have a hip fracture each year in the UK. A broken hip (fracture) happens when an older person falls. The person may have osteoporosis. People who break a hip are at risk of another fracture or of dying. The government recommends the use of specialist fracture liaison services which aim to diagnose hip fracture patients who have osteoporosis and to prescribe treatment. It is unclear how effective these services are in preventing further fractures and reducing mortality.

We used data collected by hospitals and general practitioner surgeries to look at changes in death rates and second fractures after a hospital starts a new service or changes an existing service. We identified dates when changes were made by speaking with doctors and nurses working at each hospital studied.

The appointment of a specialist doctor or nurse working in fracture prevention was associated with a reduction in death rates in patients following a hip fracture. Specialist doctors had a bigger effect than nurses. There was no effect on the chance of patients going on to have a second hip fracture. After working out the costs of these services to the NHS, appointing a specialist doctor is more cost-effective than employing a specialist nurse.

Interviews with doctors and nurses working in these services suggested that little is done to follow up patients after they leave hospital. Many patients stop taking their medication and do not participate in exercise programmes or other effective treatments. This may be a reason why services do not reduce the chances of a second hip fracture.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 11/1023/01. The contractual start date was in May 2012. The final report began editorial review in July 2015 and was accepted for publication in November 2015. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2016. This work was produced by Judge et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Services and Delivery Research Editor-in-Chief

Professor Jo Rycroft-Malone Professor of Health Services and Implementation Research, Bangor University, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Professor Aileen Clarke Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor Elaine McColl Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Health and Wellbeing Research and Development Group, University of Winchester, UK

Professor John Norrie Health Services Research Unit, University of Aberdeen, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk